Author's response to reviews

Title: Prediction of uterine dehiscence using ultrasonographic parameters of cesarean section scar in the nonpregnant uterus: a prospective observational study.

Authors:

Michal Pomorski (pomorski.md@gmail.com)
Tomasz Fuchs (tfuchs@o2.pl)
Mariusz Zimmer (mzimmer@op.pl)

Version: 2 Date: 27 May 2014

Author's response to reviews:

Dear Journal Editorial Office Team

As requested line and page numbers are included.
The first page of the manuscript is the title page. It includes names, affiliation, email addresses and full postal address of the submitting author.

Michal Pomorski, MD PhD.